

Responses to the reviewers:

Dear Editor (Dr. Yuan Qi) and respectful reviewers (no; 02438752, 02441672, 00009064, 03317125)

Greetings,

I highly appreciate for your time spent to improve the paper in to the actual international paper and I agree with all your comments. Back in to the details of all comments arrived, I conclude that there is no idea about scientific parts of the manuscript, however, I need to revise the paper with the help of an English native person to improve the English proof.

I asked one of my friends who is native English speaker to help me. Here you can have revised version according her comments.

Best wishes

Amin

**Corresponding author:**

Amin Talebi Bezmin Abadi (PhD)

Assistant Professor,

Department of Bacteriology, Faculty of Medical Sciences, Tarbiat Modares University, Tehran, Iran

Address: Tarbiat Modares University, Department of Bacteriology, P.O. Box: 14115-111, Tehran, Iran

Office: +98 21 82884883

Fax: +98 21 82884883

Lab: +98 21 82884803